Title: Evaluation of effect of low-level laser therapy on adolescents with temporomandibular disorder

Authors:

Camila HL Godoy (cami_godoy@outlook.com)
Paula FC Silva (paulafernanda84@yahoo.com.br)
Deise S Araujo (deise.fisio@ymail.com)
Lara J Motta (larajmotta@terra.com.br)
Daniela A Bisotto-Gonzalez (dani_atm@uninove.br)
Fabiano Politti (politti@uninove.br)
Raquel A Mesquita-Ferrari (raquel.mesquita@gmail.com)
Kristianne PS Fernandes (krisporta@gmail.com)
Regiane A Carvalho (regianealbertini@uninove.br)
Sandra K Bussadori (sandra.skb@gmail.com)

Version: 2 Date: 7 May 2013

Author's response to reviews:

Dear editorial board,
I´m sending attached the revised manuscript, with the trial registration number and with the table at the end of the manuscript.
It’s also attached the grant form (I translated the begining of the 1st page to make it easier to understand) and the approval from the Human Ethics Commitee as requested.
If you any other document, just let me knnow.
Your truly,
Camila Godoy